Author's response to reviews

Title: Geriatric assessment in undergraduate geriatric education - a structured interpretation guide improves the quantity and accuracy of the results: a cohort comparison

Authors:

Tobias Deutsch (tobias.deutsch@medizin.uni-leipzig.de)
Elisabeth Igenbergs (e.igenbergs@web.de)
Thomas Frese (thomas.frese@medizin.uni-leipzig.de)
Hagen Sandholzer (hagen.sandholzer@medizin.uni-leipzig.de)

Version: 2 Date: 22 August 2012

Author's response to reviews:

I've made the revisions according to your suggestions: The investigation was performed in accordance with the declaration of Helsinki and the ethics commitee of the Leipzig Medical School stated that there is no approval necessary.